Author's response to reviews

**Title:** A global approach to the management of EMR (Electronic Medical Records) of patients with HIV/AIDS in Sub-Saharan Africa: the experience of DREAM Software

**Authors:**

Andrea Nucita (a.nucita@gmail.com)  
Giuseppe M Bernava (massimo.bernava@gmail.com)  
Michelangelo Bartolo (michelangelo.bartolo@fastwebnet.it)  
Fabio Di Pane Masi (fabiodipane@gmail.com)  
Pietro Giglio (piero.giglio@gmail.com)  
Marco Peroni (peronimail@yahoo.it)  
Giovanni Pizzimenti (giovanni.pizzimenti@unime.it)  
Leonardo Palombi (lepalom@tin.it)

**Version:** 4  **Date:** 26 August 2009

**Author's response to reviews:** see over
Changes in the last version of Manuscript 2060524122245701 (August 25, 2009)

a) Affiliation modified to match criteria  
b) “Fig 1” changed in “Figure 1” on page 8  
c) Tables with standard numbers instead of roman numerals  
d) Removed vertical lines from tables